Author's response to reviews

Title: Characterizing pre-dialysis care in the era of eGFR reporting: a cohort study

Authors:

Khaled Abdel-Kader (abdelkaderk@upmc.edu)
Gary S Fischer (fischerg@upmc.edu)
James R Johnston (jamiej@pitt.edu)
Chen Gu (guc@upmc.edu)
Charity G Moore (moorecg@upmc.edu)
Mark L Unruh (unruh@pitt.edu)

Version: 6 Date: 10 March 2011

Author's response to reviews:

Dear Editors,

I have altered the color of the text in Figure 1. The previous red text was only meant to highlight the changes made in response to reviewer's comments. All of the text in the figure is now black. I have also reviewed the remainder of the manuscript and figures, and all other portions of the article are acceptable.

Many thanks,
Khaled